Author’s response to reviews

Title: Water extract from processed Polygonum multiflorum modulate gut microbiota and glucose metabolism on insulin resistant rats

Authors:

Wen Gu (guwen1230@qq.com)
Min Yang (1330542876@qq.com)
Qian Bi (biqian920125@163.com)
Lin-Xi Zeng (1125478751@qq.com)
Xi Wang (371154787@qq.com)
Jin-Cai Dong (1909242258@qq.com)
Feng-Jiao Li (2564630932@qq.com)
Xing-Xin Yang (yxx78945@163.com)
Jing-Ping Li (a349231148@qq.com)
Jie Yu (cz.yujie@gmail.com)

Version: 1 Date: 18 Jun 2019

Author’s response to reviews:

Dear who it may concerned:

Hello.

I have revised my original manuscript, and provided more literature to prove that insulin resistance may be related to dyslipidemia and pathoglycemia, and PPM (processed Polygonum multiflorum) is widely used in the treatment of dyslipidemia and dysglycemia and the like.

On the other hand, our experimental results also prove that we successfully established a rat model of insulin resistance, and PPM had therapeutic effects on diet-induced IR in rats.

The format of the reference has also been updated to ensure that it meets the system requirements.
We appreciate your consideration of our manuscript, and look forward to hearing from you at your earliest convenience.